Over-education and Gender Occupational Differences in Spain

Abstract This paper explores the role of over-education in shaping the negative relationship between the education level attained by employees and the fact of working in a gender-dominated occupation, in Spain, a country where the phenomenon of over-education is common. Applying multinomial logit regressions, and controlling for individual and job characteristics, the results confirm the typical finding that having a university degree decreases the odds of working in a gender-dominated occupation. However, this is only true in the case of women when considering long - more than three years - university studies. The evidence also suggests that the general spread of over-education in Spain weakens that relationship so that reducing over-education would eventually lead to more uniformity in the gender-distribution of employment across occupations.

Keywords Over-education, Occupations, Gender differences, Human capital.

1. Introduction

The increase in female employment in developed countries, spurred by increases in services and the marketization of women's housework and care-work, has facilitated a concentration of working women in specific occupations related to their "innate abilities" or "competitive advantages", such as the service and clerical sectors. This, in turn, has led to a surge in gender-inequality distribution across occupations and in gender-occupational segregation. In many European countries, this process has been favoured by the public sector, allowing the outsourcing of nurturing and care and, by providing family-friendly working conditions, enabling the reconciliation of paid work and family responsibilities (Costa 2000; Emerek et al. 2001).

Within Europe, north-south differences in the extent of marketization of women's housework and care-work, and in increases in female employment, has resulted in a differential evolution of the process across countries. Thus, there is today a broad dispersion within European countries of women's participation in the labour market and in gender job segregation. Emerek et al. (2001) and Bettio (2002) show that, for the last decades of the twentieth century, both these variables were positively correlated, so that countries with higher rates of female employment also exhibited greater gender segregation. In this context, segregation may be interpreted as the price paid by women in exchange for participation in the European labour markets.

In recent decades, however, the level of gender occupational segregation has shown a downward trend in most developed countries (Anker 1998; Costa 2000). Despite female employment continuing to rise, increases in both the education level attained by women and in the development of new technologies have made it possible for women to enter traditionally male employment and participate in new jobs created, thus reducing occupational segregation (Blau et al. 1998; Black and Juhn 2000; Dolado et al. 2001). However, this does not appear to be the case in Spain, since the increases observed in education levels and in female employment have been accompanied by a steady increase in gender segregation, from 1994 to 2008 (Otero and Gradín 2001; Cebrián and Moreno 2008; Iglesias and Llorente 2010).

Against this background, we test the hypothesis that a mismatch between educational attainment and job requirements may be shaping the association between higher education and lower segregation. Given that Spain is one of the OECD countries

2

with the highest levels of over-education (in excess of 25%; OECD 2009), this may be a reasonable argument for explaining the persistence of gender occupational segregation over time. The existing literature has paid little attention to factors affecting gender segregation in Spain, and much of that research has focused on technical issues related to the measurement of segregation, while empirical studies have centred on assessing its magnitude and its evolution over time. Among the latter, there is evidence at the aggregate level showing certain correlations between personal characteristics and segregation (Dolado et al. 2001, 2004; Mora and Ruiz-Castillo 2003; Cebrián and Moreno 2008; Iglesias and Llorente 2010; del Rio and Alonso-Villar 2010, 2012). A common finding in these studies is that a higher level of education is correlated with a lower level of segregation. However, a systematic and rigorous econometric analysis of the factors influencing gender segregation, with individual data, has not yet been carried out, nor has the presence of significant over-education been taken into account.

In this paper, our aim is to investigate the relationship between personal and jobrelated characteristics, and gender inequality in occupational distribution in Spain. As explained below, we do not specifically aim to assess the impact of such characteristics on a measure of gender segregation; rather, we are interested in the link between those characteristics and the (unequal) distribution of men and women across occupations, with our main contribution being the special check for educational variables, including over-education. In pursuit of that, we carry out multivariate regression analyses using cross-sectional data to find robust associations between the personal and labor characteristics and the fact that certain occupations are gender-dominated.1

Our results suggest that the general belief that a more educated working population helps to reduce gender-inequality in the distribution of workers across occupations appears to apply in the Spanish case, with our main finding being that this relationship would be stronger if over-education were less extensive. Additionally, we find that short university studies, of three years of less, largely achieved by women in certain segregated fields such as health and teaching, may generate a more unequal gender distribution of workers across occupations. In this respect, the paper has two major

¹ A better knowledge of the relationship between individual characteristics and gender-inequality distribution is essential in reducing labor segmentation and wage differences across genders, since occupational segregation is viewed as one important source of the gender wage gap (Bayard et al. 2003; Amuedo-Dorantes and de la Rica 2006). The unavailability of wage information in the database we use prevents an analysis of the influence of occupational segregation on the gender wage gap.

policy implications. First, to establish measures that are not solely focused on stimulating the acquisition of higher levels of qualification, but are also designed to achieve a better matching between educational attainment and job requirements. In this line, lifelong learning activities may be used as a way to compensate for an initial mismatch. Second, to adopt measures encouraging more equal gender distribution in the choice of field of study, and to promote policies aimed at removing those norms or stereotypes that exogenously assign jobs and occupations to workers of one specific gender

The structure of the paper is as follows. Section 2 presents a brief summary of the literature and Section 3 provides statistical information about participation, segregation, and the evolution of human capital variables in Spain. In Section 4, we develop the empirical model that serves to relate these factors to the observed gender occupation differences in the Spanish labour market. Section 5 shows our estimated results, and Section 6 concludes.

2 Literature review and descriptive information

Most traditional explanations for gender-occupational inequality are based on three main arguments.² First, human capital accumulation (education, experience, and tenure) may differ across gender so that individuals search for jobs that are more suited to their characteristics (Mincer and Polachek 1974).³ Second, workers choose specific jobs according to their preferences (England 2010) and to the characteristics of those jobs (e.g. women may wish to work in jobs that allow them to balance work and family responsibilities, Becker 1985), but also because the cultural norms and stereotypes in society lead them to follow "traditional" gender roles (Bergmann 1974). Third, discrimination against women exists (Becker 1985), and only those firms/sectors that do not discriminate accept women.⁴ More recently, the literature has also focused on the fact that the job satisfaction of women increases with the number of women in a

² For reviews of theoretical and empirical findings, see Anker (1997) and Bettio and Verashchagina (2009). A succinct review of the topic can be seen in European Commission (2009).

Alternatively, it has been argued (e.g., Bergmann 1995) that, provided there is segregation in the labour market, men and women choose the human capital level that is most appropriate for them.

There may also be statistical discrimination, when firms suspect that women are more likely to interrupt their careers due to maternity and child rearing, and so assign them to jobs demanding lower human capital accumulation (Coate and Loury 1993).

particular occupation. Thus, psychological considerations, such as social identity theories, may help to explain gender-occupational inequality. Akerlof and Kranton (2000) claim that individuals may suffer a loss of utility when working in jobs with which they do not "identify". In this framework, social norms or stereotypes may make individuals happier working with those of the same gender. Similarly, Alesina and La Ferrara (2000) show that the individual utility derived from joining a group increases with the share of own-kind group members, and decreases with the share of a different kind.

The topic of over-education has been extensively studied in the literature, with emphasis on the negative effects of returns and on job satisfaction and on the phenomena of the bumping down and/or crowding out of less skilled workers (Moen 1999; Hartog 2000). Over-education is derived from a growth in the supply of highereducated workers that exceeds the growth in demand. This may arise from individuals' choices in order to compensate for a lack in other observed and unobserved skills (Groot and Maasen van den Brink 2000), and as a way of gaining access to employment or to search for a better job (Dolado et al. 2009). Regarding differences between men and women, Groot and Maasen van den Brink (2000) find that women who have experienced a career interruption, e.g. due to child rearing, are more likely to be in jobs for which they are over-educated. Frank (1978) argue that, when women are the second earner in the household and their decisions are subordinated to those of men, they may suffer from restricted mobility and are more prone to be in a job for which they are over-educated.

The empirical literature in Spain has reported partial findings that we take as starting points in our contribution. To our knowledge, no study has focused on the relationship between over-education and the gender distribution in employment across occupational categories. The bulk of the literature on gender disparities in the Spanish labor market has focused on wage discrimination (Gardeazábal and Ugidos 2005; Amuedo-Dorantes and De la Rica 2006). The analysis of segregation has centred either on its measurement (Mora and Ruiz-Castillo 2003; Alonso-Villar and Del Rio 2010a) or on its evolution over time (Cebrián and Moreno 2008; Iglesias and Llorente 2010). There exist studies that search, to some extent, for correlations between segregation indices and certain worker characteristics. For example, Mora and Ruiz-Castillo (2003) show that more educated and older workers are associated with lower gender segregation. Dolado et al.

5

(2001, 2004) compare the US with EU countries, including Spain, finding that occupational segregation is lower among the most educated, especially if they are young. These studies also find that part-time work is related to higher segregation. Ibañez-Pascual (2008) applies multinomial logit techniques to estimate the probability of being in a gender-dominated occupation, depending on a set of individual and job characteristics, finding that education helps to reduce segregation, as do firm characteristics such as public ownership, with discrimination and stereotypes playing an important role in shaping gender differences across occupations.

Somewhat different results are presented in Del Rio and Alonso-Villar (2010, 2012) and Alonso-Villar and del Rio (2010a). These authors suggest a complementary approach to traditional "overall" measures of segregation, in order to explore the segregation of target population subgroups (women or blacks, for example). Whereas overall measures, such as the Dissimilarity Index, ID (Duncan and Duncan, 1955) and IP (Karmel and MacLachlan, 1998), inter alia, compare the distribution of two population subgroups one to the other, these authors propose local measures of segregation by which the distribution of one subgroup (women or men) is compared with the distribution of total employment in the economy, and not with the distribution of the other group. These studies axiomatically develop several local measures, showing that they possess good properties and connecting with overall segregation measures (Alonso-Villar and del Río 2010b; del Río and Alonso-Villar 2010). These measures are then used in studying the Spanish case, investigating, in turn, segregation by gender and age, by gender and educational attainment, and by gender and types of contract. 5 The authors find, first, that segregation is greater among women than among men, indicating that women are more unevenly distributed across occupations than men. Second, they show that greater segregation is observed for the youngest (16-29 years old), the lowest educated, and those with temporary contracts (in the case of men), or those with parttime contracts (in the case of women). Segregation is also greater for those in low-paid jobs. Specifically, in terms of education levels, they conclude that a higher education level is not associated with lower segregation, since segregation among the mediumeducated (upper secondary) is lower than among the highly-educated (university and

⁵The case of the distribution across occupations of natives and immigrants in Spain is examined in del Rio and Alonso-Villar (2012) and Alonso-Villar and del Rio (2013).

similar), and the fact that, despite sample women exhibiting higher average education levels than men, the latter are less segregated.6

The issue of measurement of gender segregation is progressively growing in extension and in scope.7 Also, there is a growing literature on the relationship between group characteristics (education, for example) and the observed level of segregation.8 However, as argued by Gradín (2013), the procedures followed do not allow for multiple attributes, or are only appropriate when dealing with small units, which is not appropriate in our case. In consequence, we do not attempt to study the relationship between education and a particular measure of segregation. Rather, our aim focuses on examining the correlation between the characteristics capturing education and the unevenness of gender distribution across occupational categories, using multinomial logit techniques.

3 Descriptive information

In this Section, we briefly review the evolution of employment in Spain in recent decades, and present a battery of statistics showing the importance of gender inequality distribution in Spain and its relationship to education variables. Until the onset of the Great Recession, Spain enjoyed a very large increase in total employment, of which female employment represented more than half.

3.1 The increase in Spanish female employment until 2007

Our data come from the Spanish Labor Force Survey (LFS, Encuesta de Población Activa), conducted by the Spanish Institute of Statistics (INE), following Eurostat guidelines. This survey provides labor market information for a representative sample of Spanish households. We use the period 1986 to 2010 to study the evolution over time of employment indicators, focusing on year 2010 for a picture of the more recent

⁶ The authors argue that this may result from some kind of segregation in the choice of university studies, since women choose to enroll in fields where their future career prospects are better, such as health or teaching (del Rio and Alonso-Villar 2012).

Measures of randomness based on exposure indices (Hellerstein and Neumark 2008; Aslund and Skans 2009) arise in contraposition to measures of evenness (ID, IP, and Gini, Atkinson and Theil indices). Among these latter, measures of local segregation have been proposed as complementary to measures of "overall" segregation (Alonso-Villar and del Río 2010b; del Río and Alonso-Villar 2010).

Some studies link the results of multinomial logit regressions with ID-type indices of occupational segregation (Spriggs and Williams 1996; Kalter 2000). Some others compute segregation of specific subgroups of the population (Hellerstein and Neumark 2008; Aslund and Skans 2009).

situation as well as for estimation purposes. Occupations are considered at a two-digit level of the Spanish National Classification of Occupations (CNO-1994) in this section, while it is disaggregated at the three-digit level in the estimations section.9 These lists comprise 66 and 206 occupations, respectively.

Between 1986 and 2007, around nine million jobs were created in Spain, of which five-and-a-half million were for women. As a consequence, female participation and female employment rates both increased by around 20 percentage points (p.p.) in that period (see Table 1). These figures have come closer to the EU averages, but are still below the norm, 2 and 5 p.p., respectively. Similarly, these figures have been approaching male rates, but differences in 2007 were still substantial, around 20 p.p. in favour of men.¹⁰ The growth in female employment has centred on the younger, the married, and the more educated. A generational or cohort effect has been witnessed, with women progressively entering the labour market, and remaining in it for a longer period than some decades ago (Cebrian and Moreno 2008). More than 90% of the increase in Spanish female employment between 1986 and 2007 was allocated to services.

(Insert Table 1)

3.2 A descriptive approach to inequality

We now focus on the distribution of working men and women across occupational categories. In order to save space, while still providing insightful evidence, we focus on those occupations that employ the largest numbers of both men and women in the Spanish labour market. Thus, we take the ten occupations for which the ratio between the number of working men in each occupation and total male employment is the highest; and we do the same for the case of women. This ratio is known as the concentration index (see Tables 2a and 2b). Whereas the ten most female-centric occupations represent more than 63% of total female employment (more than 50% with

⁹ CNO Clasificación Nacional de Ocupaciones (National Classification of Occupations) is based on the 1988 ISCO classifications, but they do not entirely coincide. Recently, the 2011 CNO tries to better adjust to the 2008 ISCO, but the data utilised in this study fits only the older classification. In our Tables, the full description of occupations is presented in order to clarify which jobs are included.

During the Great Recession, gender differences have declined due to the great loss of jobs, especially among men. Thus, in 2010 (see the last column in Table 1), the male participation rate is "only" 16 p.p. above that of the female rate, with the male employment rate being 13 p.p. higher than that of females. More dramatically, male and female unemployment rates have basically converged to a 25% value in the last few years, compared to a decade ago when the difference was about 10 p.p.

only the six most concentrated occupations), this figure is considerably lower among men (barely 44%), revealing a greater homogeneity in the concentration of men across occupations, as standard deviations confirm.¹¹ Men work largely in manufacturing and construction (machinery and related trades workers, assemblers, labourers), about 41% of total male employment, and also in certain services (drivers, science and engineering professionals, and financial associate professionals), less than 53%, whereas female employment is primarily concentrated in the services sector (more than 85% of total female employment).

Column 4 shows the representation ratios of those occupations, computed as the ratio of female (male) share of employment in each occupation to female (male) share of total employment. Ratios above 1 indicate female (male) over-representation in an occupation, with ratios below 1 indicating under-representation. In 2010, the share of females (males) in total employment was 46% (54%). Looking at the female case, the representation ratios for the ten occupations analysed are always higher than 1.2. The position in the ordering of those occupations with a larger share of women is displayed in parenthesis, confirming that the vast majority of workers in those female-centric occupations are women. In the case of men, the evidence is somewhat different, since there are occupations with representation ratios below 1. Additionally, only five of the ten occupations considered rank among the first 15 occupations according to the share of men.¹²

(Insert Tables 2a, 2b)

Following Hakim (1993) and Anker (1998), we construct a classification of occupations into three categories, depending on the share of men (women) in each occupation. An occupation is labelled as masculinised (male-dominated, or men over-represented) if in this occupation the share of men is above the national proportion of males in employment (54%), plus a threshold (e.g. 20%). Similarly, an occupation is said to be feminised (female-dominated, or women over-represented) if the share of women in that occupation is above the national proportion of females in employment (46%), plus the same threshold. The occupations in between are said to be integrated, a category that includes those occupations with a share of women between 26% and 66%.

¹¹ For similar results in Spain, see Alonso-Villar and del Rio (2010a) and del Rio and Alonso-Villar (2010). This finding is also typical in Europe (Eurostat 2008; European Commission 2009).

In this respect, there are many other occupations for which the replacement ratios for men are also high, above 1.5, but their contribution to total employment is less significant.

According to this criterion, eight out of the ten most female-centric occupations are labelled as feminised, whereas six out of the ten most male-centric occupations are masculinised. Thus, Table 3 shows that, whereas the weight in total employment is quite balanced across the three types of occupations, only 11% of workers in the masculinised occupations are women, and only 21% in the feminised occupations are men. This criterion can also be used to describe the evolution of gender distribution across occupations and over time. Thus, in 2002, eleven occupations were feminised, increasing to twelve in 2010. Similarly, masculinised occupations rose from twenty four in 2002 to twenty eight in 2010, whereas integrated occupations declined from thirty one to twenty six.

One second approach to check for the increase in gender differences across occupations is with the most widely-known indicator of gender segregation, the Index of Dissimilarity (Duncan and Duncan 1955). This index is computed as $ID = 1/2\sum_{i} |m_i - f_i|_{W} \text{ here } m_i(f_i) \text{ is the proportion of employed males (females) working}$

in occupation i over total male (female) employment. It ranges between 0 and 1, with 0 indicating null segregation, and 1 full segregation. With data from the Spanish LFS, the ID rose from 0.50 in 1994 to 0.55 in 2008.13 After that, the ID trends downward,

basically due to the effects of the Great Recession, which have caused a high level of unemployment in Spain, especially among men in male-dominated occupations as, for example, in construction.¹⁴ The ID for 2010 reached a value of 0.53.

(Insert Table 3 here)

3.3 Gender occupational inequality and educational variables

Once it has been shown that the most concentrated occupations are also among the most unequally gender-distributed, we focus on the characteristics related to educational variables in those occupations. Generally speaking, many of the most feminised and

¹³Many studies have analysed the evolution of the ID in Spain (Otero and Gradin 2001; Mora and Ruiz-Castillo 2003; Iglesias and Llorente 2010, among others). Following the classification by Blau et al. (1998), these authors have shown that the "sex composition effect" clearly dominated the "occupation mix effect" during the 1990s, indicating that the increase in ID was largely due to the allocation of men/women into already masculinised/feminised occupations. In the late 1990s and early 2000s, this process has been accompanied by an increase in the number of feminised and masculinised occupations.

Immigration has also contributed to the increase in ID until 2007. Amuedo-Dorantes and de la Rica (2011), del Rio and Alonso-Villar (2012) and Alonso-Villar and del Rio (2013) show that immigrant women in Spain have been employed mostly in occupations where native women were more heavily concentrated.

masculinised occupations can be considered as requiring low skills. In Tables 2a and 2b, columns five to eight, the proportions of low-educated, intermediate-educated and higheducated workers are shown. The low-educated are those who have not finished secondary school; intermediate-educated, those who have completed secondary school, but no higher; and high-educated, those who have a college degree (either short or long studies).15 National distribution, shown at the bottom of Tables 2a and 2b, reveals that more than a half of the employed have only attained low (compulsory) education, whereas around one fourth have graduated from college. Thus, except for occupations related to teaching, associate professionals, and managers (which, by their very nature, require workers with a degree), the remaining occupations under consideration present quite low educational averages, with this being more evident in male-dominated occupations. This suggests that higher education may be associated with lower gender inequality. To check for this, we again use the widely-known ID that, although it is not additively decomposable (Mora and Ruiz-Castillo 2003), and hence not appropriate for subgroups, allows comparison with prior studies. Thus, the ID for those with no tertiary education is over 0.60, whereas it is less than 0.40 among the high educated.¹⁶

With data from the Spanish LFS, the proportion of the population over age 16 with above-compulsory schooling increased, between 1985 and 2007, from less than 13% to about 43%. This was accompanied by an increase in the phenomenon of over-education (Hartog 2000; Budría and Moro-Egido 2008; Ortiz 2010), confirming that the growth of higher-educated workers outpaced the growth in demand. Consequently, not all of the increase in educational attainment by the Spanish active population has led to an increase in the allocation of employees to jobs requiring higher levels of qualification. Rather, it seems that a large portion of workers may hold skills exceeding those required in their job. In this context, it is not unlikely to expect that the role of education in relieving gender inequality may be attenuated; that is, the existence of over-education may be diminishing the association between high education and lower inequality. If so,

¹⁵ From an international perspective, low-educated are ISCED2 and intermediate-educated are ISCED3. Short high studies are ISCED5 including advanced vocational schooling and university studies requiring three years at most (Diplomaturas, Títulos Técnicos). Finally, long university degrees are ISCED6, including those requiring four or more years of study (Licenciaturas, Grados) and graduate studies (Masters, Ph.D.).

These values resemble those in Dolado et al. (2004), who find that, for year 1999, and occupations interacted with activities, for those with tertiary education the ID was 0.37 in the case of younger workers, increasing to 0.49 in the case of those aged between 45 and 54 years old. These values were comparatively lower for those with less than tertiary studies, over 0.50 in all age ranges.

this may be of great importance in Spain, a country characterised by an elevated level of educational mismatch, among the highest in the OECD countries (OECD 2009; Ghignoni and Verashchagina 2013; Verhaest and Van der Velden 2013).

Column 9 in tables 2a and 2b provide a broad picture of the phenomenon in our data. Due to the way in which educational level is presented in the Spanish LFS, we have computed educational mismatch from a statistical point of view. The literature classifies four typical ways of defining educational mismatch: two are considered as subjective and two as objective (Groot and Maassen van den Brink 2000; Hartog 2000). Subjective measures come from worker self-assessment. A direct measure corresponds to the answer to the question whether the individual feels over- or under-educated for the work they do, whereas an indirect measure is obtained by comparing the actual education level of the workers with the self-report on the required education level in their job. Although there is some reason to fear upward bias in these measurements, it is often the best available measure (Hartog 2000).

The objective definitions are also of two types. The first is based on a comparison between the actual education level and the job-level requirements, established from an evaluation by professional job analysts. Conceptually, this is the best measure, but it is often difficult to implement and keep up-to-date, due to high costs and frequent changes in occupation classification, along with the escalation of job-requirements, and of the skills necessary to fill them. The second objective measure of over-education is obtained by comparing years of education attained with the average education level within the occupation of the worker. An individual is said to be over-educated if the educational level is higher than the average educational attainment in that occupation, plus a standard deviation.¹⁷ The choice of statistical as opposed to other measures of over-education is dictated by data availability. In their meta-analytic review, Groot and Maassen van den Brink (2000) find that this statistical definition yields the lowest estimate of over-education, so that our measure of over-education provides a lower bound for the phenomenon of over-education in Spain. Looking at Tables 2a and 2b, the

¹⁷Alternatively, over-education can be measured by taking the modal value as a reference, rather than the average. However, this measure has the disadvantage that when the occupational disaggregation is high, as in our study, in many occupations the modal value is barely representative. For studies measuring overeducation in Spain under different definitions, see Budria and Moro-Egido (2008) and Ortiz (2010), using the 1994-2001 European Community Household Panel; Nieto and Ramos (2010), using the EADA 2007 (the Spanish application of the European Adult Education Survey), and García and Montuenga (2012), who use the 2007-2010 waves of the Spanish Quality of Working Life Survey. The typical measure of over-education is more than 25% of Spanish employees.

problem of over-education is substantial in several occupations, especially in the case of those in low-skilled occupations, and of those in male-dominated occupations (see also Table 3).18

4. The empirical model

Our aim is to examine the relationship between characteristics capturing education and the unevenness of gender distribution across occupational categories, once we control for other human-capital-related variables, such as experience (proxied by age) or tenure, and a group of job characteristics, such as the type of contract, the sector of activity, the nationality, and the region of the worker, with the influence of over-education being specifically addressed. We carry out multinominal logit regressions in which the factor change in the likelihood of being in a gender-dominated occupation, over being in an integrated occupation, as a consequence of attaining a particular education level, and the influence of educational mismatch, are both investigated. Multinomial logit regressions allow us to compute the change in odds between several possible alternatives, depending on the characteristics and on the parameter vectors associated with those alternatives (Wooldridge 2002: 497-8; Long and Freese 2006: 224-8).

Of special interest in our investigation is evidence of how the factor change in odds of being in a gender-dominated occupation - resulting from the education level – varies when considering over-education. We are aware that, in a partial analysis, simultaneity may be at work. However, since we are using cross-section data, it is difficult to control for this with instrumental variables or longitudinal data. As a consequence, significant coefficients must be interpreted merely as partial correlation, not necessarily reflecting causality. Our guess is that the association between education and gender occupational distribution may be negatively affected when workers are employed in jobs requiring lower qualifications than those they actually hold. This may be of singular magnitude in a country like Spain, where over-education is among the most significant in the OECD countries.

Table 4 provides the definitions of our variables of interest and their average values, with data from the Spanish LFS for year 2010, from more than 60,000 observations

¹⁸Educational attainment is provided in the Spanish LFS in eight levels, which are used to compute the incidence of over-education. However, for purposes of description and estimation, we group them into the four levels described in Table 4.

corresponding to wage-earners only. There are more men (54%) in the sample than women (46%), and more than a half of the individuals live in couple. Around 80% of our sample are in the age-range 25-54 years old, and 30% have tenure of between 4 and 10 years (29% with shorter tenure and 39% longer). In order to ascertain the individual significance of each education level, we have four categories:19 those who have attained compulsory studies at most (representing about 55% of the total employed); those with non-compulsory secondary studies (20%); those with short higher studies (11%); and those with upper university degrees (14%). Some 6% of individuals are immigrants, 20% work in the public sector, 76% are on a permanent contract, and 14% work parttime. Only 20% of the surveyed individuals work in Information and Communication Technology sectors (high-ICT sectors).20 We find that almost 25% of the workers are considered to be over-educated. By gender, women are younger, less tenured, more heavily concentrated in part-time jobs, and more frequently occupied in the public sector and in high-ICT sectors, than men. Women are also less over-educated and on average more educated than men. Noticeably, the percentage of women with short university studies is double that of men.

(Insert Table 4)

5. Estimated results

We investigate the relationship between human-capital and job-related variables, and gender distribution, by considering three types of occupations. Following the classification discussed above, we distinguish between male-dominated, female-dominated, and integrated occupations.²¹ In 2010, 35% of occupations are male-dominated, against 33% female-dominated, and 32% integrated. We carry out a

¹⁹As said in the previous footnote, information is provided at a more disaggregated level. We finally decided to combine diverse primary and compulsory secondary studies under the label of compulsory studies, since preliminary analyses reveal no statistically significant differences between these education levels in the relationship with unevenness of gender distribution.

We follow Iglesias-Fernández et al. (2011) who consider as ICT sectors only those sectors in the Knowledge Industry, which includes the following manufacturing and service activities (codes corresponding to the ISIC 3.1 Rev from the OECD): 22 Publishing, printing and reproduction of recorded media; 30 Manufacture of office, accounting and computing machinery; 32 Manufacture of radio, television and communication equipment and apparatus; 64 Post and telecommunications; 72 Computer and related activities; 73 Research and Development; 80 Education, and 92 Recreational, cultural and sporting activities. Technological development helps to reduce both gender segregation (Black and Juhn 2000; Dolado et al. 2001) and over-education (Ghignoni and Verashchagina 2013).

In constructing the dependent variables, occupations are taken at the 3-digit 1994 CNO level to diminish the possible influence of vertical segregation.

multinomial logit regression to analyse the factors associated with working in a genderdominated (male or female) occupation, relative to working in an integrated occupation, which is taken as the referent group. In tables, we present the relative risk ratios, or odds ratios, which, given that all variables are of binary type, express how the relative risk of being in a masculinised (feminised) occupation changes as a consequence of a status change in an independent variable, given that the other variables in the model are held constant.22

Results of our baseline model for year 2010 are shown in Table 5. Values above 1 indicate that the likelihood of being in a gender-dominated occupation, relative to being in an integrated occupation, increases after a unit change in the corresponding independent variable (other variables remaining constant), whereas values below 1 indicate a decrease. The first column in each pair of estimations refers to the change in the likelihood of working in a male-dominated occupation, relative to an integrated occupation, and the second column to the change in the likelihood of working in a female-dominated occupation, also relative to working in an integrated occupation. The first pair of columns reports estimates without the dummy variable indicating whether the individual is considered to be over-educated, and in the second pair of columns estimates are obtained when that dummy variable is included. As shown in Table 5, this dummy is found to be statistically significant, revealing the importance of explicitly considering a measure of educational mismatch in investigating the association between the unequal gender-distribution of workers across occupations and educational attainment. Nine sector fixed effects are included, to take into account that the same occupation may have different gender distributions across industries. Regional fixed effects are added, to account for certain cultural and stereotype differences across the 17 Spanish Autonomous Communities (NUTS 2 regions), as well as other time-constant determinants.23 The referent categories are: women aged 35 to 44 years, not living in couple, with non-compulsory secondary studies, in a job for which they are not overeducated, with tenure between 4 and 10 years. These women are native, working fulltime on a temporary contract, in a non-high-technology industry in the private sector.

(Insert Table 5)

²²Relative risk ratios are computed, as exponentials, from the multinomial logit estimated coefficients (see Long and Freese 2008: 178).

These estimates are not reported, but are available upon request. In Tables, significant odds ratios are marked with asterisks.

In what follows, when interpreting the odds for each of the dependent variables, the remaining variables are held constant, and the odds are computed relative to the referent category, being in an integrated occupation. (We refrain from adding these phrases in order to facilitate the reading of the results.) Values statistically significant above 1 may be interpreted as variables associated with greater inequality across occupations, whereas values statistically significant below 1 can be interpreted as variables associated with greater equality across occupations. Looking at the first two columns in Table 5, we observe that the odds of being in a male-dominated occupation are seven times greater for men than for women. Analogously, for men, the odds of being in a female-dominated occupation decrease by a factor 0.27. Provided that a large array of characteristics is controlled for, the gender coefficient can be interpreted as a measure of the odds of women/men with similar characteristics working in a gender-dominated occupation. Estimates are very different from 1, suggesting that the observed characteristics do not totally explain the unequal allocation of men and women into occupations. Thus, aside from the existence of discrimination, unobserved characteristics – such as field of study, preferences, stereotypes, etc...- represent potential arguments for the observed gender differences in occupations, suggesting that identity theories from psychological perspectives (Akerlof and Kranton 2000; Alesina and La Ferrara 2000) may also help in understanding the existence of such differences.

The odds of being in a gender-dominated occupation are expected to increase in the case of those living in couple (by a factor close to 1.1) relative to those who do not. Regarding the age variables, the most obvious result is that for younger workers (under 25), the odds of being in a male-dominated occupation are clearly lower than 1, which may be interpreted as recent cohorts of workers being associated with lower inequality. However, this does not hold for the case of being in female-dominated occupations, for which the odds are usually greater than 1 and statistically significant (save for those over 55). The category of reference when measuring education is non-compulsory secondary studies, for which, by construction, the odds are 1. There are clear differences between male- and female-dominated occupations. Thus, university education is expected to decrease the odds of being in a male-dominated occupation by factors of 0.85 and 0.42, respectively. However, only the case of having a degree of four or more years of university studies is expected to decrease the odds of being in a male-dominated occupation by factors of 0.85 and 0.42, respectively. However, only the case of having a degree of four or more years of university studies is expected to decrease the odds of being in a male-dominated occupation by factors of 0.85 and 0.42, respectively. However, only the case of having a degree of four or more years of university studies is expected to decrease the odds of being in a female-dominated occupation; the fact of having a degree of three or fewer years of university

16

studies does increase the odds, rather than reducing them. One possible explanation for this result is that occupations in which this kind of study (intermediate university) is required (associate professionals in social science, in the sciences, in biomedical, or teaching in general) are nests for women, who are mostly allocated into these feminised occupations (Tables 2a, 2b and 3).

The likelihood of being in a gender-dominated occupation is progressively reduced as tenure increases. Provided that the results express correlations and not causality, this may be interpreted in two ways: workers in integrated occupations may have lower incentives to move to other, more gender-unequal occupations, and remain in integrated occupations; but also that workers in more unevenly-distributed occupations may wish to move more often to more integrated occupations, thus reducing their seniority. For immigrants, relative to natives, the odds of being in a gender-dominated occupation are expected to increase (by factors of 1.2 and 1.3 respectively). The odds of being in a female-dominated occupation decrease by a factor of 0.71 for those working in the public sector, relative to those working in the private sector. Working part-time is expected to increase, by a factor of 1.43, the odds of working in a female-dominated occupation, and to decrease, by a factor of 0.55, the odds of working in a maledominated occupation. Similarly, working on a permanent contract is expected to increase, by a factor of 1.2, the odds of working in a female-dominated occupation, and to decrease, by a factor of 0.9, the odds of working in a male-dominated occupation. Finally, working in a high-ICT sector is expected to increase the odds of being in a male-dominated occupation, and to decrease the odds of being in a female-dominated occupation. This result will be analyzed in depth below.

Taken as a whole, a high education level, aside from gender, is the most outstanding characteristic associated with lower odds of being in an unevenly-gender-distributed occupation. This broad conclusion is even clearer when we include a dummy controlling for whether the individual is over-educated (as defined above). The last two columns show estimated results when adding this dummy, whose relative risk ratio is found to be statistically significant. Thus, for the over-educated, the odds of being in a male-dominated occupation are expected to increase by a factor of 2, whereas the odds of being in a female-dominated occupation are expected to decrease by a factor of 0.77. Thus, over-education is associated with a greater likelihood of being in a male-dominated occupation, and with a lower likelihood of being in a female-dominated

occupation. Regarding the rest of the variables, results are very similar to those shown in the first two columns, except for the education variables. These latter deserve some consideration.

First, for those having only compulsory studies, the odds of being in a maledominated occupation increase by a factor of 1.57 (as against 1.22, when over-education is not controlled for), whereas for those having a university degree, the odds of being in a male-dominated occupation decrease considerably (a factor of 0.51 for those with degrees of three or fewer years, as against 0.85 without the over-education dummy; and by a factor of 0.19 for those with longer-term degrees, as against 0.42, without the overeducation dummy). Second, in the case of female-dominated occupations (relative to integrated occupations) the changes in the odds are less marked, with respect to the case without the over-education dummy. The main difference is in those having attained an education level corresponding to university studies of four or more years, for which being in a female-dominated occupation is associated with a decrease in the odds of 0.14, as against 0.42 when the over-education variable is not included. Again, educational attainment of a shorter university degree is associated with greater odds of being in a female-dominated occupation, by a factor of 1.28.

All these results suggest that university studies are associated with lower inequality in gender distribution across occupations. Our results coincide with most of the existing literature for the Spanish case - although they are counter to those of del Rio and Alonso-Villar (2010, 2012). Our argument in support of this distinct result is that those authors show that women's segregation is greater, despite that women are more educated, but, while they do show that women's segregation fell between 1994 and 2009, they do not relate this to the general increase in educational attainment. Additionally, controlling for over-education, as we do, makes this evidence more clearcut. Thus, we contribute to the literature by showing different behaviour across higher education levels, and focusing on the relevance of over-education. In the case of femaledominated occupations, possession of a four-year degree appears to be the minimum level needed before we find a positive association between high education and low inequality. The inclusion of the over-education dummy shows that factor changes in the odds could have been higher if the educational attainment and the job requirements were appropriately matched. These results lead us to two main findings, put simply as follows. First, a higher education level appears to be positively associated with nongender-dominated occupations. Second, despite the fact that over-education is associated with lower odds of being in a female-dominated occupation, the high education-low inequality link would be stronger if over-education were not present. This latter is of special relevance in the Spanish case, where over-education is not uncommon.24

Although educational attainment in Spain has increased in recent decades, the question remains whether the positive relationship found between educational attainment and gender equality distribution across occupations is homogenously observed across all age ranges, or whether there is a certain age range for which the relationship is stronger. A second question that requires some attention is the extent to which our results are robust when potentially different family situations are taken into account. Labour-market participation and occupational choice may be associated with household composition and/or the number of dependent individuals in the household. Ideally, knowing the number of individuals in the household - as well as their labour status - would help us to control for this, but the Spanish LFS does not provide that information. Alternatively, we use other information provided in the dataset relative to i) whether individuals live in couple, or not; and ii) whether one or more individuals in the household are employed. An additional factor that we consider worthy of analysis is participation (or not) in occupations with an elevated component of ICT. Theoretically, technological change promotes reductions in inequality by allowing women to enter traditional male occupations (Black and Juhn 2000; Dolado et al. 2001, 2004). Specifically, ICT-based occupations should be more evenly distributed, since this kind of work is not affected by past patterns of segregation, and women's participation is expected to be higher; such work requires less physical aptitude, more easily accommodates family and work duties, and vacancies are more abundant.25

To investigate these hypotheses, we re-run multinomial logit regressions focused on these aspects. To address the first, three different subgroups of the age range are considered. We concentrate on the middle-aged group, to avoid distortion from the periods of high education and of pre-retirement, and distinguish between ages 25-34,

²⁴The different behavior observed between degrees of three or fewer years, and degrees of four or more, may underlie the finding in the series of papers by Alonso-Villar and del Río on the greater segregation of women. If a significant proportion of graduate women are allocated to occupations where short higher studies are required, such a result is not unexpected.

The increase in employment in ICT occupations has helped to reduce the gender wage gap in Spain, but gender inequality in occupations has not been reduced (Iglesias-Fernandez et al. 2011).

35-44, and 45-54. Regarding the second question, we analyze two different cases: i) individuals living in couple (married or cohabitant), relative to individuals who do not (single, separated, divorced, or widowed); and ii) individuals in a household being the sole employee, relative to those in households in which two or more people are working. With respect to the final aspect, we distinguish between working in ICT sectors, or not.

Table 6 displays the odds corresponding to the three different age ranges defined above. (In order to save space, only the values for education level and type of sector are presented.) The values present a similar character to those presented in the last two columns of Table 5 (note that the over-education variable is included in estimates presented in Table 6). In general terms, for those with university studies of longer than three years, the odds of being in gender-dominated occupations are expected to largely decrease, whereas for those with only compulsory studies, the odds of being in male-dominated occupations. For those with short university studies, the odds of being in female-dominated occupations are expected to increase slightly. Similar to the case shown in Table 5, being over-educated, or working in a high-ICT sector, is expected to increase the odds of being in a male-dominated occupation, and decrease the odds of being in a female-dominated occupation.

(Insert Table 6)

Across age ranges, the most outstanding changes can be seen in the "senior" workers, age range 45-54 years. First, university degrees of fewer than four years study are associated with increases in the odds of being in female-dominated occupations, by a factor of 1.5 (much higher than the value reported in Table 5 for all ranges, 1.28). In the other two age ranges, differences in the odds between being in female-dominated occupations, and being in integrated occupations, are not statistically significant. Second, the decrease in the odds of being in gender-dominated occupations, as a consequence of attaining the highest education level, is greatest among the senior workers. Third, for the over-educated, as the age range increases, the odds of being in male-dominated occupations are expected to increase more, and the odds of being in female-dominated occupations increase less. Therefore, younger over-educated workers are expected to suffer from lower gender-inequality distribution across occupations than

20

older over-educated workers. Fourth, for those working in a high-ICT sector, the odds of being in a female-dominated occupation are expected to fall, whereas the odds of being in a male-dominated occupation are expected to increase, especially in the case of younger workers.

Considering the age ranges as proxies for cohort effects, these results suggest that the Spanish labour market is dynamic and that the entry of younger cohorts is helping to redesign its structure. Our results show that, in the lower age ranges, short university studies are no longer associated with greater unequal-gender-distribution across occupations. This may be due to the fact that traditionally female-dominated occupations requiring such levels of education, as for example those related to nursing, teaching, and the like, have progressively given way to the entry of male workers during the last decades. Similarly, the most extreme values observed in the group of workers aged 45-54, referring to long university studies or over-education, may be merely indicating the low proportion of highly-educated (and consequently, of over-educated) workers in this group. Finally, results indicate that the expansion of employment in high-ICT sectors observed in recent years in Spain (see Iglesias et al. 2011) is positively associated with being in male-dominated occupations, but not in female-dominated occupations.

Table 7 shows the odds for the two cases considered: living in couple, or not, and being the sole employee in the household, or not. Estimates for each of the subgroups mimic the general behaviour shown in the last two columns in Table 5, with little or no difference between them. One minor variation we might note is that having a short-cycle university degree is expected to increase the odds of being in a female-dominated occupation by factors of 1.4 when the individual does not live in couple, and 1.5 when only one member of the household is employed. Note that these factors are significantly above the average value of 1.3.

Table 8 contains the results of the relative risk ratios when distinguishing between high-ICT sectors and non-high-ICT sectors. Two major differences across education and sectors are worth mentioning. First, having only a compulsory education is expected to increase the odds of being in a male-dominated occupation only for those in non-high-ICT sectors. Second, shorter university studies are associated with increases in the odds of being in a female-dominated occupation, again only for those working in non-high-ICT sectors. Taken together, it is apparent that higher education is related to a more equal gender distribution of workers across occupations, in those sectors with high ICT. For over-educated workers, the odds of working in a gender-dominated occupation are expected to increase, by factors greater than 2, in the high-ICT sectors. By contrast, for the over-educated in non-high-ICT sectors, the odds of working in a male-dominated occupation are expected to increase, and the odds of working in a female-dominated occupation are expected to decrease (mirroring the results in Table 5). The general view is that the direct association of higher education and more equal gender distribution across occupations is more evident in high-ICT sectors, supporting the notion of developing ICT sectors as a way to reduce inequality, with over-education dampening that relationship.

(Insert Tables 7 and 8)

6 Conclusions

The aim of this article is to investigate the association, at the micro level, between a variety of personal and job characteristics of workers, focusing on the significant role of over-education, and inequality in the distribution of men and women across occupational categories. We classify occupations into three categories: male-dominated, where men are over-represented; female-dominated, where women are over-represented; and integrated, where the share of women is roughly similar to that of men. In the descriptive section of the paper, we show that female-dominated occupations are largely populated by women (80%), whereas male-dominated occupations are almost entirely filled by men (90%). We also show, first, that many of those working in gender-dominated occupations have only attained the lowest educational level and, notwithstanding that, the phenomenon of over-education is quite common in these occupations.

Our multivariate analysis, based on multinomial logit regressions using individual data, provides the main result of our study: aside from gender itself, which may capture unobserved characteristics, such as field of study, preferences, stereotypes, etc., or the existence of discrimination, the educational attainment of the worker is the most prominent factor associated with the odds of working in an integrated occupation. In particular, university education is strongly related to decreases in the odds of working in a gender-dominated occupation, with this association being more apparent once overeducation is controlled for. Thus, our results suggest that the general belief of a more

educated working population helping to reduce gender-inequality in the distribution of workers across occupations, appears to characterise the Spanish case, and that this relationship would be stronger if over-education were less extensive. In this respect, an immediate recommendation for policy is to establish measures that are not only focused on stimulating the acquisition of higher levels of qualification,26 but are also designed to achieve a better matching between educational attainment and job requirements. Consequently, policies stimulating lifelong training, and the appropriate choice of professional careers, appear to be of singular importance. This would eventually help to reduce cross-gender wage differentials, since occupational segregation is viewed as one important source of the gender wage gap in Spain (Amuedo-Dorantes and de la Rica 2006).

A second important result is that the relationship between higher education and greater gender-equality differs between the two types of gender-dominated occupations. University degrees with duration shorter than four years are associated with unevenness in gender distribution only in female-dominated occupations, whereas in maledominated occupations, these degrees are associated with greater gender equality. This result suggests that short university studies, which are largely pursued by women in certain specific fields (e.g. Health and Teaching) have generated more unequal distribution, but have also allowed women to enter the labour market, especially in services and in the public sector. This calls for the adoption of measures pursuing more equal gender distribution in the choice of studies, and for policies aimed at removing those norms or stereotypes that exogenously assign jobs and occupations to workers of one specific gender. This process appears to be already at work, since when analysing the relationship between education and gender inequality across age ranges, which may proxy for cohort effects, we see that the evidence seems to be weaker for new entrants to the labour market. Thus, in young cohorts (those aged 25-34 years), with short university degrees, the odds of working in male-dominated occupations decrease, and the odds of working in female-dominated occupations do not increase. We may then conclude that changes in cultural patterns and in labour, tax, and social legislation have coincided with the increase in education levels, to generate better conditions for the

²⁶Spain today has one of the highest shares, among OECD countries, of the adult population with only compulsory education, close to 50% (OECD 2009).

extension of gender equality. Family situation appears to play only a minor role in shaping the association between education and gender-occupational distribution.

Finally, we have studied the effect of working in sectors with a significant influence of ITC, in the relationship between education level and gender inequality in occupations. Our first finding is that working in these sectors increases the odds of being in a male-dominated occupation (relative to integrated occupations), but decreases the odds of being in a female-dominated occupation (also relative to integrated occupations). This result can be better understood when looking at different age ranges, where we see that the evidence is more clear-cut in the younger generation, and when looking at different ICT sectors, since over-education has a larger presence in high-ICT sectors. Thus, it can be interpreted that the fact of working in sectors with high participation in ICT is associated with working in gender-dominated occupations, due to the development of these sectors in recent decades, and the general extension of overeducation. In this context, it appears that the development of ICTs should be accompanied by an adequate match between job requirements and worker capabilities to reduce gender inequality.

This is a static analysis that could be improved upon by analysing flows across jobs, to investigate whether labour mobility helps to reduce inequality. This, and the fact that a structural analysis is not possible with the existing data (and in consequence we cannot talk about causality, but only about correlations), leaves the door open for further research on this topic. With the results obtained in this present study, an improvement in the match between jobs and employees is of undoubted importance in decreasing undesirable inequality in the distribution of men and women across occupations.

Acknowledgements

We thank participants at the 7th Annual International Conference (ATINER) in Athens, to two anonymous referees and the Editor of this Journal for helpful comments and suggestions. We acknowledge the financial aid provided by the Spanish Women's Institute, code 190; Bank of Santander and University of Zaragoza project UZ2012-SOC-03; and the Autonomous Government of Aragón (Consolidated Research Group S13).

24

References

- Akerlof, G., & Kranton, R. (2000). Economics and identity. Quarterly Journal of Economics, 115(3), 715-752.
- Alesina, A., & La Ferrara, E. (2000). Participation in heterogeneous communities. Quarterly Journal of Economics, 115, 847-904.
- Alonso-Villar, O., & del Río, C. (2010a). Segregation of female and male workers in Spain: Occupations and industries. Hacienda Pública Española / Revista de Economía Pública, 194 (3), 91-121.
- Alonso-Villar, O., & del Río, C. (2010b). Local versus overall segregation measures.Mathematical Social Sciences, 60(1), 30-38.
- Alonso-Villar, O., & del Río, C. (2013). Occupational segregation in a country of recent mass immigration: Evidence from Spain. The Annals of Regional Science, 50(1), 109-134.
- Amuedo-Dorantes, C., & de la Rica, S. (2006). The role of segregation and pay structure on the gender wage gap: Evidence from matched employer-employee data for Spain. B.E. Journals in Economics Analysis and Policy: Contributions to Economic Analysis & Policy, 5(1), 1-32.
- Amuedo-Dorantes, C., & de la Rica, S. (2011). Complements or substitutes? Task specialization by gender and nativity in Spain. Labour Economics, 18, 697-707.
- Anker, R. (1997). Theories of occupational by sex: An overview. International Labor Review, 136(3), 315-339.
- Anker, R. (1998). Gender and jobs: Sex segregation of occupations in the world.Geneva: International Labor Office Press.
- Aslund, O., & Skans, O. (2009). How to measure segregation conditional on the distribution of covariates. Journal of Population Economics, 22, 971-981.
- Bayard, K., Hellerstein, J., Neumark, D., & Troske, R. (2003). New evidence on sex segregation and sex differences in wages from matched employee–employer data. Journal of Labor Economics, 21(4), 887–922.
- Becker, G. (1985). Human capital, effort, and the sexual division of labor. Journal of Labor Economics, 3, 533-558.
- Bergmann, B. (1974). Occupational segregation, wages and profits when employers discriminate by race or sex. Eastern Economic Journal, 1, 103-110.

- Bergmann, B. (1995). Becker's theory of the family: Preposterous conclusions. Feminist Economics, 1(1), 141-150.
- Bettio, F. (2002). The pros and cons of occupational gender segregation in Europe.Canadian Public Policy, 28, 65-84.
- Bettio, F. & Verashchagina. A. (2009). Frontiers in the economics of gender. London: Routledge.
- Black, S. & Juhn. C. (2000). The rise of female professionals: Are women responding to skill demand? The American Economic Review, 902, 450-455.
- Blau, F, Simpson, P., & Anderson, D. (1998). Continuing progress? Trends in occupational segregation in the United States over the 1970s and 1980s. Feminist Economics, 4(3), 29-71.
- Budría, S., & Moro-Egido, A. (2008). Education, over-education, and wage inequality:Evidence for Spain, Economics of Education Review, 27, 332–341
- Cebrián, I., & Moreno, G. (2008). La situación de las mujeres en el mercado de trabajo español. Desajustes y retos. Economía Industrial, 367, 121-37.
- Coate, S., & Loury, G. (1993). Will affirmative action policies eliminate negative stereotypes? American Economic Review, 85(5), 1220–1240.
- Costa, D. (2000). From mill town to board room: The rise of women's paid labor. Journal of Economic Perspectives, 14(4), 101-122.
- Del Río, C., & Alonso-Villar, O. (2010). Gender segregation in the Spanish labor market: An alternative approach. Social Indicators Research, 98(2), 337-362.
- Del Río, C., & Alonso-Villar, O. (2012). Occupational segregation of immigrant women in Spain. Feminist Economics, 18(2), 91-123.
- Dolado, J., Felgueroso, F., & Jimeno, J. (2001). Female employment and occupational changes in the 1990s: How is the EU performing relative to the US? European Economic Review, 45, 875-889.
- Dolado, J., Felgueroso, F., & Jimeno, J. F. (2004). Where do women work? Analysing patterns in occupational segregation by gender. Annals D'economie et de Statistique, 71-72, 293-315.
- Dolado, J. J., Jansen, M., & Jimeno, J. F. (2009). On the job-search in a matching model with heterogeneous jobs and workers. Economic Journal, 119, 200–228.
- Duncan, O., & Duncan, B. (1955). A Methodological analysis of segregation indexes.American Sociological Review, 20, 210-217.

- Emerek, R., Figueiredo, H., González, P., Gonas, L., & Rubery, J. (2001). Indicators on gender segregation. Report for the European Commission's Expert Group on Gender and Employment (EGGE). European Commission, Bruxelles.
- England, P. (2010). The gender revolutions. Uneven and stalled. Gender and Society, 24, 149-166.
- European Commission. (2009). Gender segregation in the labor market. Root causes, implications and policy responses in the EU. European Commission's Expert Group on Gender and Employment (EGGE). Publications Office of the European Union, Luxembourg.
- Eurostat. (2008). The life of women and men in Europe. A statistical portrait. Publications Office of the European Union, Luxembourg.
- Frank, R. H. (1978). Why women earn less: the theory and estimation of differential overqualification. American Economic Review, 68(3), 360–373.
- García, I. & Montuenga, V. (2012). El capital humano en la empresa aragonesa. Consejo Económico y Social de Aragón, Zaragoza (Spain).
- Gardeazabal, J. & Ugidos, A. (2005). Gender wage discrimination at quantiles. Journal of Population Economics, 181, 165-179.
- Ghignoni, E. & Verashchagina, A. (2013). Educational qualifications mismatch in Europe. Is it demand or supply driven? Journal of Comparative Economics, http://dx.doi.org/10.1016/j.jce.2013.06.006
- Gradín, C. (2013). Conditional occupational segregation of minorities in the U.S. Journal of Economic Inequality, forthcoming.
- Groot, W., & Maassen van den Brink, H. (2000). Over-education in the labor market: A metaanalysis. Economics of Education Review, 19, 149-158.
- Hakim, C. (1993). Segregated and integrated occupations: A new approach to analysing social change. European Sociological Review, 9(3), 289-314.
- Hartog, J. (2000). Over-education and earnings: Where are we, where should we go? Economics of Education Review, 19, 131-147.
- Hellerstein, J., & Neumark, D. (2008). Workplace segregation in the United States: Race, ethnicity, and skill. Review of Economics and Statistics, 90(3), 459-477.
- Ibáñez-Pascual, M. (2008). La segregación ocupacional por sexo a examen. Características personales, de los puestos y de las empresas asociadas a las

ocupaciones masculinas y femeninas. Revista Española de Investigaciones Sociológicas (Reis), 123, 87-122.

- Iglesias, C., & Llorente, R. (2010). Evolución reciente de la segregación laboral por género en España. Revista Universitaria de Ciencias del Trabajo, 11: 81-105.
- Iglesias-Fernández, C., Llorente-Heras, R., & Dueñas-Fernández, D. (2011). ICT employment, over-education and gender in Spain. Do information and communication technologies improve the female labor situation? New Technology, Work and Employment, 25, 238-252.
- Kalter, F. (2000). Measuring segregation and controlling for independent variables.Manheimer Zentrum f
 ür Europa
 üsche Sozialforschung.
- Karmel, T., & MacLachlan, M. (1988). Occupational sex segregation: Increasing or decreasing? Economic Record, 64, 187-195.
- Long, S., & Freese, J. (2006). Regression models for categorical dependent variables using Stata. Texas: Stata Press.
- Mincer, J., & Polachek, S. (1974). Family investment in human capital: Earnings of women. Journal of Political Economy, 82(II), 76-108.
- Moen, E.R. (1999). Education, ranking, and competition for jobs. Journal of Labor Economics, 17, 694-723.
- Mora, R., & Ruiz-Castillo, J. (2003). Additively decomposable segregation indexes. The case of gender segregation by occupations in Spain. Journal of Economic Inequality, 12, 147-179.
- Nieto, S., & Ramos, R. (2010). Sobreeducación, educación no formal y salarios: Evidencia para España, Documento de Trabajo FUNCAS, 577.
- OECD. 2009. Education at a glance 2009 OECD indicators. Paris: OECD.
- Ortiz, L. (2010). Not the right job, but secure one: over-education and temporary employment in France, Italy and Spain. Work, Employment and Society, 24, 47–64.
- Otero, S., & Gradín, C. (2001). Segregación ocupacional en España, una perspectiva territorial. Hacienda Pública Española, 159(4), 163-190.
- Spriggs, W., & Williams, R. (1996). A logit decomposition analysis of occupational segregation: Results for the 1970s and 1980s. Review of Economics and Statistics, 78(2), 348-355.
- Verhaest, D., & van der Velden, R. (2013). Cross-country differences in graduate overeducation and its persistence. European Sociological Review, 29(3), 642-653. 28

Wooldridge, J. (2002). Econometric analysis of cross section and panel data. Cambridge, MA and London: The MIT Press.

Table 1 Employment indices in Spain (several years)

	· · · · · ·	1986			1989			1994			2000			2007			2010	
Rates	М	W	Т	М	W	Т	М	W	Т	М	W	Т	М	W	Т	М	W	Т
Participation	69.7	30.0	48.7	68.6	33.7	50.6	65.7	37.1	51.0	66.5	41.4	53.4	70.2	49.4	59.0	68.1	52.3	60.1
Employment	57.2	21.8	38.5	59.8	25.0	41.9	52.9	25.3	38.7	60.1	32.9	46.1	65.1	43.8	54.0	54.7	41.6	48.0
Unemployment	18.0	27.4	21.0	12.8	25.7	17.2	19.6	31.6	24.1	9.5	20.4	13.9	6.4	10.8	8.3	19.8	21.0	20.0

M, W, T stand for men, women and total, respectively. Data come from the Spanish Labor Force Survey (Encuesta de Población Activa), Spanish Institute of Statistics (INE).

	Number of employed (in thousands)		FemaleCompulsory Representationnon-un Ratio (ranking)compu	iversity univers	sity educated			
Women								
Domestic cleaners and helpers in offices, hotels and other establishments (91)	1,186	14.61	1.99 (1)	86	10	2	2	33
Personal services workers (51)	830	10.22	1.90 (2)	53	39	6	2	9
Sales workers (53)	676	8.33	1.51 (10)	69	22	5	4	8
Administrative, specialised secretaries, and associate professionals in finance,	509	6.27	1.48 (12)	33	29	20	18	38
mathematics, and regulatory government (34)								
Cooks, waiters, and bartenders (50)	500	6.16	1.20 (15)	78	17	3	2	44
Secretaries and assistant clerks (with customer services tasks) (44)	369	4.55	1.66 (4)	42	33	14	11	23
Primary school teachers and related professions (28)	332	4.10	1.63 (5)	2	4	75	19	18
Secretaries and assistant clerks (without customer services tasks) (43)	264	3.25	1.58 (6)	44	30	14	12	25
University and higher education teachers and related professions (22)	250	3.09	1.21 (14)	0	0	16	84	15
Teachers in first university cycle in natural sciences and related professions (27)	197	2.42	1.77 (3)	0	0	96	4	4
Standard deviation (coefficient of variation)	3.82 (.60)							
National average	е			55	20	11	14	25

Table 2a Occupational distribution by gender, average education levels, and average educational match: the 10 most female-centric occupations. Year 2010

Data come from the Spanish Labor Force Survey (Encuesta **de Población Activa**), Spanish Institute of Statistics (INE). Occupations are defined at the two-digit CNO 1994 classification (in parentheses are the corresponding codes for 66 occupations). Female Concentration Index computed as the percentage of women in an occupation over total employed women. Female Representation Ratio computed as the ratio of female share of employment in each occupation to female share of total employment (46%), order in the ranking is included in parentheses. Educational attainment is the percentage of female workers with over-educated: percentage of female workers with over-education: an individual is said to be over-educated if the educational level is higher than the average educational attainment in that occupation, plus a standard deviation.

	Number of employed (in thousands)	Male Concentration Index	MaleCompuls Representationno		ShortLongOver- niversity educated			
	ulousalius)	muex	Ratio (ranking)co	mpulsory dear				
Men			runne (running)ee	inpuisory degr	eesdegrees			
Drivers (86)	678	6.63	1.79 (10)	81	17	1	1	36
House builders and related trades workers (71)	601	5.88	1.83 (1)	87	11	1	1	26
Building finishers, painters, and related trades workers (72)	544	5.33	1.80 (6)	67	31	1	1	44
Machinery mechanics and repairers; electrical and electronic trades	477	4.67	1.82 (4)	51	46	2	1	59
workers (76)								
Business and administrative associate professionals (33)	412	4.03	1.28 (29)	48	22	13	17	36
Cooks, waiters, and bartenders (50)	407	3.98	0.83 (51)	78	17	3	2	23
Physical, chemical, engineering and ICT (information and	389	3.81	1.40 (25)	20	48	17	15	32
communication technicians) associate professionals (30)								
Protective services workers (52)	346	3.38	1.70 (13)	65	23	6	6	12
Managers in firms with 10 or more workers (11)	315	3.08	1.37 (26)	28	12	19	41	46
Sales workers (53)	295	2.89	0.56 (56)	70	22	5	3	8
Standard deviation (coefficient of variation)	1.24 (.28)							
National avera	ge			55	20	11	14	25

Table 2b Occupational distribution by gender, average education levels, and average educational match: the 10 most male-centric occupations. Year 2010

Data come from the Spanish Labor Force Survey (Encuesta **de Población Activa**), Spanish Institute of Statistics (INE). Occupations are defined at the two-digit CNO 1994 classification (in parentheses are the corresponding codes for 66 occupations). Male Concentration Index computed as the percentage of men in an occupation over total employed men. Male Representation Ratio computed as the ratio of male share of employment in each occupation to male share of total employment (46%), order in the ranking is included in parentheses. Educational attainment is the percentage of male workers with over-educated: percentage of male workers with over-educated if the educational level is higher than the average educational attainment in that occupation, plus a standard deviation.

Number of occupations	Occupation type	Proportion in employment	Men	Computeroryer- universityedu	Secondary non- cat ed mpulsory	Short university degree	
	1			degree 728Mala26	896522632	10	
				726Integra	ted325552221926		
				2912Female	332146151021		
Data come from the	e Spanish Labor Ford	e Survey (Encuesta d	e Poblac	ión Activa), Spar	ish Institute of Stati	stics (INE). Occupations are defined at the	
two-digit CNO 199	4 classification (66 c	ccupations). Proportio	on in emp	loyment is the per	centage of occupation	ons of each type over total number of	
occupations (66) N	Aen is the nercentage	of male workers in ea	ch type c	foccupation Edu	cational attainment i	s the percentage of workers in each	

Table 3 Descriptive statistics by groups of occupations (percentage). Year 2010

occupations (66). Men is the percentage of male workers in each type of occupation. Educational attainment is the percentage of workers in each educational level (definition in Table 4). Over-educated: percentage of workers with over-education: an individual is said to be over-educated if the educational level is higher than the average educational attainment in that occupation, plus a standard deviation.

Variable name	Definition	Sample Average	Male	Female
Personal characteristicss				
Gender1, male; 0, female		0.54		
Marital status	1, live in couple (married/cohabitant); 0, not (single/separated/divorced/widowed)	0.57	0.60	0.55
Age				
< 25	Under 25 years	0.06	0.06	0.06
25-34	Between 25 and 34 years	0.23	0.21	0.25
35-44	Between 35 and 44 years	0.29	0.29	0.29
45-54	Between 45 and 54 years	0.27	0.27	0.27
> 54	Over 54 years	0.15	0.16	0.13
Educational level				
Compulsory studies	Primary and compulsory secondary studies	0.55	0.59	0.49
Secondary studies	Non-compulsory secondary studies	0.20	0.20	0.20
University 1	High vocational training. Degrees corresponding to 3 or less years of university studies	0.11	0.08	0.16
University 2	Degrees corresponding to 4 or more years of university studies	0.14	0.13	0.16
Overeducation	When the individual holds an educational level that is higher than the average educational attainment in her occupation, plus a standard deviation	0.25	0.27	0.22
Tenure				
Tenure < 1	Less than one year	0.14	0.13	0.15
Tenure 1-3	Between 1 and 3 years	0.15	0.13	0.18
Гenure 4-10	Between 4 and 10 years	0.29	0.27	0.32
Fenure 11-20	Between 11 and 20 years	0.20	0.22	0.19
Tenure 21-30	Between 21 and 30 years	0.13	0.15	0.11
Γenure > 30	More than 30 years	0.06	0.10	0.06
mmigrant	Immigrant=1; Native=0	0.06	0.06	0.07
Job characteristics				
Public sector	1, public sector; 0, private sector	0.20	0.16	0.24
Part-time	1, part-time; 0, full-time	0.14	0.05	0.23
Type of contract	1, permanent; 0, temporary	0.76	0.78	0.74
Sector of technology		-		
High-ICT Sectors	Knowledge Industry (see footnote 20)	0.12	0.10	0.15
Number of observations		65,292	36,032	29,260

Table 4 Definition and average values of relevant variable	s. Year 2010
--	--------------

Data come from the Spanish Labor Force Survey (Encuesta **de Población Activa**), Spanish Institute of Statistics (INE). Occupations are defined at the three-digit CNO 1994 classification (206 categories).

 Table 5 Multinomial logit model. Relative Risk Ratios of being in a male- or femaledominated occupation, relative to being in an integrated occupation. Year 2010

> Female-Male-dominateddominatedMale-dominated Female-dominated

```
Male7.56***0.27***7.38***0.28***
Marital status1.11***1.05*1.12***1.05*
16-240.79***1.15***0.80***1.13***
25-340.971.11**0.981.10***
45-540.951.15***0.961.15***
> 541.021.051.051.04
Compulsory studies1.22***1.001.57***0.97
University 10.85***1.17***0.51***1.28***
University 20.42***0.42***0.19***0.14***
Overeducation2.00***0.77***
Tenure < 11.22***1.18***1.20***1.17***
Tenure 1-31.14***1.17***1.15***1.17***
Tenure 11-200.80***0.88***0.82***0.88***
Tenure 21-300.62***0.82***0.64***0.82***
Tenure > 300.47***0.890.48***0.87*
Immigrant1.17***1.30***1.021.39***
Public sector0.980.72***1.000.71***
Part-time0.55***1.45***0.52***1.46***
Type of contract0.90***1.19***0.91***1.18***
High Tech sector1.16***0.66***1.30***0.63***
Log likelihood-37939-37597
  2LR χ (p-value)39824 (0.0000)40510 (0.0000)
     2
Pseudo R0.34420.3501
Observ.52,882
All regressions include dummies for nine occupations (1-digit level), nine industries, and the 17 NUTS II
Spanish regions. *, **, *** indicate significant at 10, 5 and 1% level respectively.
```

Table 6 Multinomial logit model. Relative Risk Ratios of being in a male- or female-
dominated occupation, relative to being in an integrated occupation, by age range (only
education and sector variables). Year 2010

Age groups
25-3435-4445-54
Male-Female-Male-Female-
dominated dominated dominated dominated dominated
Compulsory studies1.49***1.011.87***1.081.45***0.94
University 10.56***1.090.50***0.970.46***1.47***
University 20.21***0.20***0.20***0.12***0.15***0.09***
Overeducation1.78***0.79***2.03***0.85***2.70***1.02
High Tech sector1.98***0.67***1.27***0.57***1.020.74***
Log likelihood-9250-11025-9695
210,160.55 (0.0000)11,757.30 (0.0000)11,070.18 (0.0000)LR χ (p-value)
2
Pseudo R0.35450.34780.3634
Observ.13,10315,53213,888
All regressions include dummies for nine occupations (1-digit level), nine industries, and the 17 NUTS II
Spanish regions, in addition to the rest of regressors in Table 5. *, **, *** indicate significant at 10, 5 and 1% level respectively.

Table 7 Multinomial logit model. Relative Risk Ratios of being in a male- or female- dominatedoccupation, relative to being in an integrated occupation, by family situation (only educationvariables). Year 2010.

Live in coupleLive aloneOne employedMore employed Male-Female-Male-Female-Male-Female-Male-Female- dominated dominated dominated dominated dominated dominated dominated
Compulsory studies1.72***0.961.41***0.981.56***1.071.57***0.93
University 10.53***1.18***0.49***1.41***0.50***1.49***0.51***1.21***
University 20.18***0.11***0.21***0.18***0.19***0.16***0.19***0.13***
Overeducation2.17***0.79***1.85***0.74***1.92***0.82***2.05***0.75***
Log likelihood-21548-15842-12118-25375
224,490.26 (0.0000)16,335.95 (0.0000)13,287.38 (0.0000)27,116.86 (0.0000)LR χ (p-value)
2
Pseudo R0.36240.34020.35410.3482
Observ.31,00321,87917,40435,478
-All regressions include dummies for nine occupations (1-digit level), nine industries and the 17 NUTS II Spanish regions, in addition to the rest of regressors in Table 5. *, **, *** indicate significant at 10, 5 and 1% level respectively.

Table 8 Multinomial logit model. Relative Risk Ratios of being in a male- or female-
dominated occupation, relative to being in an integrated occupation, by level of
technology (only education variables). Year 2010.

High-tech sectorNon high-tech sector
Male-Female-Male-Female-
dominated.dominated.dominated
Compulsory studies0.981.031.6 /***0.96
University 10.46***1.090.56***2.07***
University 20.15***0.12***0.22***0.14***
Overeducation3.39***1.98***1.92***0.69***
Log likelihood-5200-30985
25297.17 (0.0000)37182.24 (0.0000)LR χ (p-value)
2
Pseudo R0.33750.3750
Observ.7,33045,552
All regressions include dummies for nine occupations (1-digit level), nine industries, and the 17 NUTS II
Spanish regions, in addition to the rest of regressors in Table 5. *, **, *** indicate significant at 10, 5 and
1% level respectively.